Author's response to reviews

Title: A national survey of services for the prevention and management of falls in the UK

Authors:

Sarah E Lamb (s.lamb@warwick.ac.uk)
Joanne D Fisher (j.d.fisher@warwick.ac.uk)
Simon Gates (s.gates@warwick.ac.uk)
Rachel Potter (r.potter@warwick.ac.uk)
Matthew W Cooke (M.W.Cooke@warwick.ac.uk)
Yvonne H Carter (yvonne.carter@warwick.ac.uk)

Version: 2 Date: 28 May 2008

Author's response to reviews: see over
27th May 2008

The Editorial Office
BMC Health Services Research Journal

Services for the prevention of falls in the UK

Dear Sir, Madam,

We were recently commissioned by the NiHR SDO in the UK to undertake a comprehensive national survey to characterise the most common type of falls pathway, and the development of services in the UK. We have undertaken a high quality survey that has almost universal coverage, and the findings have important, although somewhat unpopular implications for improving service development and provision.

Yours faithfully,

[Signature]

Professor Sallie Lamb DPhil, MSc, MCSP, SRP
Professor of Rehabilitation
Director Clinical Trials Unit
Warwick Medical School Clinical Trials Unit
The University of Warwick
Gibbet Hill Road
Coventry
CV4 7AL
Corresponding authors:

S E Lamb, DPhil, MSc, MCSP, SRP 1, 2

J D Fisher, BSc (hons), PhD 1

S Gates, PhD 1

R Potter, MSc, RGN 1

M W Cooke, PhD, MB, ChB, FRCS(Ed), FCEM, DipIMC 1

Y H Carter, MD FRCGP FMedSci 1

1 Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, Gibbet Hill Road, Coventry CV4 7AL.

2 Kadoorie Critical Care Research Centre, John Radcliffe Hospital, Oxford OX3 9DU.

Telephone and e-mail address:

S. Lamb: 024 7615 0404, s.lamb@warwick.ac.uk

J. Fisher: 024 7615 0404, j.d.fisher@warwick.ac.uk

S. Gates: 024 7615 0404, s.gates@warwick.ac.uk

R. Potter: 024 7615 0404, r.potter@warwick.ac.uk

M. Cooke: 024 7615 0404, M.W.Cooke@warwick.ac.uk

Y. Carter: 024 7615 0404, Yvonne.carter@warwick.ac.uk

Fax: 024 7657 4657